Author's response to reviews

Title: Neuropsychological effects of chronic low-dose exposure to polychlorinated biphenyls (PCBs): A cross-sectional study

Authors:

Martin Peper (peper@psychologie.uni-freiburg.de)
Martin Klett (martin.klett@rhein-neckar-kreis.de)
Rudolf Morgenstern (rudolf.morgenstern@charite.de)

Version: 5 Date: 12 October 2005

Author's response to reviews: see over
Re: Ms Martin Peper, Martin Klett, and Rudolf Morgenstern: Neuropsychological effects of chronic low-dose exposure to polychlorinated biphenyls (PCB): A cross-sectional study

Dear Dr. Grandjean,

Dear Dr. Ozonoff,

changes were done as suggested. I understand that the first affiliation should not be given twice on the title page. I therefore deleted my second address. However, the problem of the dual affiliation remains. I suggest the following solution: I deleted my second affiliation on top but added my present address to “correspondence”:

§Corresponding author; present address: University of Freiburg, Department of Psychology, Belfortstr. 20, 79085 Freiburg, Germany

I hope that this is acceptable.

In regard to Tables 2 and 4, they need to fit the journal page in portrait format. If this is not possible, they need to be uploaded as separate files.

The problem of additional files is that these will not be displayed in the final, published form of the article. Therefore, I compressed all Tables a little more and hope that they will fit the page now.

Please feel free to correct minor language errors directly in the final version. If further changes are necessary, please contact me. We hope that this version will meet with your approval.

Yours sincerely,
Martin Peper, PhD MD